A Systematic Review of Bimanual Motor Coordination in Brain–Computer Interface

Poraneepan Tantawanich[®], Chatrin Phunruangsakao[®], Shin-Ichi Izumi, and Mitsuhiro Hayashibe[®], *Senior Member, IEEE*

Abstract-Advancements in neuroscience and artificial intelligence are propelling rapid progress in braincomputer interfaces (BCIs). These developments hold significant potential for decoding motion intentions from brain signals, enabling direct control commands without reliance on conventional neural pathways. Growing interest exists in decoding bimanual motor tasks, crucial for activities of daily living. This stems from the need to restore motor function, especially in individuals with deficits. This review aims to summarize neurological advancements in bimanual BCIs, encompassing neuroimaging techniques, experimental paradigms, and analysis algorithms. Thirtysix articles were reviewed, adhering to the Preferred **Reporting Items for Systematic Reviews and Meta-Analyses** (PRISMA) guidelines. The literature search result revealed diverse experimental paradigms, protocols, and research directions, including enhancing the decoding accuracy, advancing versatile prosthesis robots, and enabling realtime applications. Notably, within BCI studies on bimanual movement coordination, a shared objective is to achieve naturalistic movement and practical applications with neurorehabilitation potential.

Index Terms—Brain–computer interface, bimanual coordination, neuroscience, machine learning.

I. INTRODUCTION

B RAIN-COMPUTER interface (BCI) systems capture and translate brain activity into artificial commands, enabling users to control peripheral devices. BCI applications address physical impairments by restoring motor functions through brain signal-controlled prostheses or orthosis, promoting neuroplasticity crucial for motor rehabilitation [1], [2]. However, most studies have focused on decoding motor tasks involving

Received 13 May 2024; revised 18 October 2024 and 11 December 2024; accepted 18 December 2024. Date of publication 25 December 2024; date of current version 9 January 2025. This work was supported by the JSPS Grant-in-Aid for Scientific Research on Innovative Areas Hyper-Adaptability Project 22H04764. (Poraneepan Tantawanich and Chatrin Phunruangsakao are co-first authors.) (Corresponding authors: Poraneepan Tantawanich; Mitsuhiro Hayashibe.)

Poraneepan Tantawanich is with the Department of Robotics, Graduate School of Engineering, Tohoku University, Sendai 980-8579, Japan (e-mail: poraneepan.tantawanich.q3@dc.tohoku.ac.jp).

Chatrin Phunruangsakao is with the Neuro-Robotics Laboratory, Graduate School of Biomedical Engineering, Tohoku University, Sendai 980-8579, Japan (e-mail: phunruangsakao.chatrin.p8@dc.tohoku.ac.jp).

Shin-Ichi Izumi is with Tsurumaki Onsen Hospital, Hadano 257-0001, Japan (e-mail: shinichi.izumi.c2@tohoku.ac.jp).

Mitsuhiro Hayashibe is with the Department of Robotics, the Graduate School of Engineering, the Neuro-Robotics Laboratory, and the Graduate School of Biomedical Engineering, Tohoku University, Sendai 980-8579, Japan (e-mail: hayashibe@tohoku.ac.jp).

Digital Object Identifier 10.1109/TNSRE.2024.3522168

different limbs, which activate distinct cortical regions [3]. Consequently, the available control signals remain limited. Numerous studies have decoded various motor tasks, including hand movements [4], [5], [6], finger movements [7], and reachand-grasp actions [8], [9] to address this challenge.

FMBS

Activities of daily living (ADL) necessitate coordinated and synchronized movements involving both hands, i.e., bimanual motor control, for effective completion. Furthermore, bimanual motor training enhances both bimanual and unimanual performance [10], [11]. Notably, bimanual coordination training strengthens interhemispheric communication pathways, promoting symmetrical brain functions through coordinated interactions between hemispheres [12], [13]. Consequently, numerous studies have investigated bimanual training, including inter-subject variability [12], optimizing task design for maximum efficiency [14], and comparing its efficacy to other training paradigms [15], [16], [17], [18]. Moreover, researchers have developed bimanual training devices such as robot-assisted tools and VR-integrated systems to enhance subject performance [19], [20].

Despite the widespread use of bimanual training in neurorehabilitation over several decades, research on BCI-based bimanual training remains limited. Integrating neuroimaging techniques into bimanual coordination training allows for progress monitoring and provides valuable insights into brain activation and neurological perspectives. Desrochers et al. [21] explored the impact of visuomotor perturbation on bimanual motor control by analyzing electroencephalography (EEG) spectral power across both high and low beta bands. Phunruangsakao et al. [13] investigated changes in alpha band EEG connectivity during unimanual and bimanual motor imagery, comparing measurements taken before and after bimanual training. In addition, brain activation differences were investigated using functional near-infrared spectroscopy (fNIRS) across varying complexities of unimanual and bimanual tasks [22]. These studies reveal brain activation differences during unimanual and bimanual tasks of varying complexities. Although these studies offered valuable insights into neural correlates of bimanual coordination, they did not satisfy the primary goal of BCI, which is to utilize neural signals to control peripheral devices.

Despite limited research on BCI-based bimanual training, existing studies have explored diverse directions and objectives. One of the most noticeable trends is the studies aiming for the highest classification accuracy of bimanual predefined

© 2024 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/ class, including leftward, rightward, and midward hand motions [23]. These research directions highlight impactful neurological features interpreted via high-performance classification models. Specifically, deep-learning models can benefit from comprehensive feature visualization using explainable artificial intelligence (xAI), enhancing the understanding of neural correlates [24], [25]. However, the practical implementation of these studies is limited by the predefined tasks within the paradigm. Hence, researchers have explored more flexible decoding systems. Instead of decoding predefined classes, some studies have focused on decoding position, velocity, and force trajectories [25], [26], [27], [28], [29], [30]. Additionally, a few investigations have extended motion analysis to 3-dimensional space, aiming for higher degrees of freedom (DOF) [31], [32], [33]. In addition to decoding bimanual tasks, relevant studies exploring bimanual training-related findings were identified. King et al. [34] investigated the impact of bimanual visual feedback on unimanual task performance. Additionally, another study assessed motor-imagery skills in pianist and non-pianist subject groups, representing varying levels of bimanual dexterity [35].

The primary objective of this systematic review was to consolidate knowledge on bimanual training in BCI, compare diverse approaches, and facilitate cross-study knowledge transfer. However, inherent biases make direct comparisons challenging, especially when contrasting experiments involving non-disabled individuals, patients, and non-human primates. Even when recruiting participants with similar criteria, differences in neuroimaging techniques must be considered, as different techniques capture different aspects of neural activities. Task paradigms, including motor execution, motor imagery, motor attempt, and task complexity, must also be carefully considered, as they activate distinct but overlapping cortical regions. Therefore, concluding which decoding model outperforms others based solely on accuracy comparisons is inappropriate. This review contributes to the scientific literature by addressing these challenges as follows:

- It provides an overview of the recent research trends and the state-of-the-art paradigms in bimanual training within the BCI field.
- It classifies and summarizes study details, encompassing participants, neuroimaging techniques, task designs, experimental paradigms, model inputs, feature extraction techniques, and decoders.
- It summarizes important neurological findings, evaluates their transferability and explores potential synergies with corresponding conclusions.
- It examines challenges and proposes future research directions within this context.

The remaining sections of this study are structured as follows: Section II outlines the procedures for identifying relevant references, covering search databases, conducting the screening process, and applying inclusion and exclusion criteria. Section III provides an in-depth analysis of the literature search results, organized by key topics such as study populations, neuroimaging techniques, experimental tasks, input and feature extraction methods, and decoding algorithms. Section IV discusses key findings, current challenges, and areas for future research. Finally, Section V concludes the study.

II. METHODS

The identification of relevant references adhered to widely recognized Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines [36], which set the standards for reporting systematic reviews within the scientific community. Figure 1 provides an overview of the guidelines. The search spanned several databases, including PubMed, IEEE Xplore, Scopus, and ISI Web of Science. It specifically targeted article titles, abstracts, and keyword sections of articles up to the search date of November 5, 2024.

During the identification phase, 1999 records from selected databases and three registers were initially included. These records underwent the automatic elimination of 998 duplicate records with exact matches using Zotero (version 7.0.9), a reference management software. Consequently, 1001 unique records remained for further screening.

Further screening focused exclusively on titles and abstracts. Studies not employing neuroimaging techniques were excluded. Additionally, studies centered on unimanual or lower limb tasks were omitted from this study. Furthermore, other non-bimanual tasks, such as linguistic, somatosensory attention, and navigation, were also excluded due to their lack of simultaneous usage of both hands. Moreover, unrelated studies, including those related to sensor development, supernumerary limbs, and teleoperation, were excluded. This screening process resulted in the removal of 868 records, including one without access, leaving only 133 records for subsequent analysis.

During eligibility assessment, full-text articles were accessed to identify relevant studies for reviewing the state of the art. The inclusion criteria during full-text screening encompassed studies involving participants performing upper-limb bimanual tasks that necessitated synchronization of both limbs and experimental paradigms designed to investigate the impact of bimanual training (e.g., bimanual feedback) and discuss neurological findings that may be beneficial for improving bimanual BCI systems. Exclusion criteria included studies that extracted features unrelated to neuroimaging techniques and those that did not entail decoding, classification, or the application of brain signals to external devices.

Finally, in the systematic review, 36 articles met the eligibility criteria. Figure 2 illustrates the distribution of these articles, with 30 from journal publications and six from conference proceedings. Table I summarizes key aspects of the eligible studies.

A. Other Studies on Bimanual BCI

During the eligibility stage, 28 bimanual BCI studies were excluded due to the absence of neurological findings or discussions related to bimanual movements. These studies primarily focused on enhancing classification performance through advanced feature extraction or decoding methods, using bimanual movements to control external devices, or comparing the impact of multimodal signal acquisition on decoding



Fig. 1. Crafted search terms for diverse databases, accompanied by the PRISMA [36] flowchart depicting article inclusion and exclusion criteria. The asterisk (*) denotes a wildcard search term.

performance. Although excluded from further review, these studies may still provide valuable insights for researchers in the field. A list of these studies is compiled in Table II, briefly summarizing their objectives, sensors used, tasks performed, and paradigms applied.

B. Existing Review

During the search, two review papers investigating bimanual coordination in BCI systems were identified. One of these papers [37], published in 2022, explores bimanual coordination within the context of motor imagery. It encompasses studies utilizing bimanual tasks with motor imagery to control

BCI systems and those focusing on cognitive, behavioral, and neurophysiological analysis underlying imagined bimanual tasks.

Another paper [3], published in 2023, discusses EEG-based motor BCIs for upper-limb movement, covering both unimanual and bimanual tasks. It provides a comprehensive overview of BCI system, including experimental paradigm design, data collection, signal preprocessing, neural activity correlation, feature extraction, movement intention decoding, and online or real application model testing.

The primary distinction between this review and existing literature lies in its comprehensive scope. Unlike prior reviews,

Ref.	Year	Population (n)	Sensors (n)	Task	Paradigm	Input	Feature ex- traction	Decoder
[58]	2013	NHP (2)	8 MEA (96); 8 MEA (48)	L/ R/ B reach- ing	ME/ MA/ AO	Firing rate	PETH	UKF/ kNN
[63]	2013	Healthy (99)	EEG (64)	L/ R/ B /F opening-closing	MI	ERDS	CSP	LDA
[46]	2013	Healthy (10)	EEG (64)	L/ R/ B/ F/ L+RF/ R+LF/ Re	MI	ERDS	CSP; Multi- GECSP; Multi- sTRCSP	SVM
[57]	2013	Healthy (22); MCS (5); UWS (9)	EEG (32)	B/ Re closing	MI	ERDS	Hjorth parameters; Brainrate; Wacker- mann; Hurst exponent; FFT; Coherence; GC; PDC; DTF; Entropy; Bhat- tacharyya distance; Correlation entropy	kNN/ SVM/ DADF
[44]	2014	Healthy (3)	fNIRS (45)	L/ R/ B/ Re ball	ME	ΔOxy - Hb	GLM	SVM
[43]	2016	Healthy (6)	EEG (26)	L/ R/ B/ Re	MI	ERDS	CSP	Stepwise re- gression
[67]	2017	Healthy (30)	MEG (4)	B finger tap- ping	ME	ERF	-	Magnitude squared coherence
[59]	2018	NHP (2)	2 ECoG (32)	L/ R/ B/ Re reach-and-press	ME	ERDS	Wavelet transform	LDA + PLS
[47]	2018	Healthy (9)	EEG (8)	L/ R/ B/ F opening-closing	MI	ERDS	-	Autoregression model
[40]	2018	Healthy (10)	MEG (114)	B opening- closing	ME/ MI	ERF	Mean ERF amplitude	SVM
[54]	2018	Healthy (10)	EEG (32)	L/ R/ B waving	ME/ MI	ERDS	CSP; CSPb	LDA
[52]	2019	Healthy (10)	EEG (30)	L/ R/ B/ F	MI	ERDS	OVR-CSP	SVM
[41]	2019	Healthy (15)	EEG (58)	L/ R/ B/ Re palmar/lateral grasp	ME	MRCP	MRCP am- plitudes	sLDA
[29]	2019	Healthy (68)	EEG (128); fMRI	L / R/ B/ Re tracking	MI	ERDS	Wavelet transform	Weighted sum of the alpha power
[64]	2020	Healthy (10)	EEG (32)	L/ R/ B/ F	MI	ERDS	CSP; FBCSP	SVM/ RVM/ SNN
[45]	2020	Healthy (8)	EEG (24)	L/ R/ B reach- ing	ME	MRCP	MRCP amplitudes; power sum	LDA/ SVM

TABLE I STUDIES ON BIMANUAL COORDINATION IN BRAIN-COMPUTER INTERFACE IDENTIFIED BY PRISMA PIPELINE

the examination of bimanual coordination in BCI is unrestricted by specific BCI paradigms (such as motor imagery, motor execution, motor attempt, and action observation) or neuroimaging techniques. Additionally, recent advancements are summarized, challenges are identified, and future research directions are proposed. This is particularly significant given

270

TABLE I (Continued.) STUDIES ON BIMANUAL COORDINATION IN BRAIN-COMPUTER INTERFACE IDENTIFIED BY PRISMA PIPELINE

Ref.	Year	Population (n)	Sensors (n)	Task	Paradigm	Input	Feature ex-	Decoder
[24]	2020	NHP (1)	2 ECoG (4x8)	L/ R/ B reach-	ME	ERDS	Wavelet	DenseNet
[35]	2020	Non-Pianist	EEG (8)	and-press L/ R	MI	ERDS	CSP	LDA
[53]	2020	(4), Flainst (4) SCI (1); SCD (1)	2 MEA (96); 2 MEA (88)	L/ R/ B reach- ing; reach-and- grasp	MA/ AO	Firing rate	-	OLE with ridge
[26]	2021	Healthy (9)	EEG (24); fNIRS (24)	B isometric contraction	ME	ERDS (EEG); HbO/HbR (fNIRS)	Hilbert transform (EEG)	cnnatt-mdn
[60]	2021	NHP (2)	2 ECoG (32)	L/ R/ B reach- and-press	ME	ERDS	Wavelet transform	DenseNet- 3D
[31]	2022	C4-C5 injury (1)	2 ECoG (64)	L/ R/ B reach- and-touch and wrist rotations	MI/ MA	ERDS	CCWT	REW- MSLM
[34]	2022	Healthy (14)	EEG (32)	L/ R rotation (B feedback)	MI	ERDS	SB-CSP	LDA
[25]	2022	Healthy (13)	EEG (64)	L/ R/ B reach- ing	ME	ERDS	-	EEGNet + LSTM
[55]	2022	Healthy (10); C4 injury (2)	EEG (32)	Dominant/ B/Re palmar/ lateral grasp	ME/ MA	MRCP	MRCP am- plitudes	sLDA/ SVM/ RF/ NBC
[51]	2022	Stroke (10); Older (10); Younger (10)	EEG (4)	L/ R/ B wrist extension/ flex-	ME/ MA/ AO	ERDS	FFT	Thresholding
[39]	2022	Tetraplegic (1)	2 MEA (96)	L/ R/ B / Re wrist/ elbow	MI	Firing rate	PETH	SVM
[23]	2023	Healthy (8)	EEG (64)	L/ R/ B reach-	ME	MRCP	-	EEGNet + BiLSTM
[65]	2023	Healthy (15)	EEG (8)	L/ R/ B/ F	MI	ERDS	ITD	FNN
[56]	2023	Healthy (18)	MEG (306)	B/ F/ WORD/ SUB	MI	ERF	Linear ker- nel	SVM/ GPC
[48]	2023	Healthy (8)	EEG (24)	L/ R/ B reach- ing	ME	MRCP; ERDS	DWT (MRCP); CWT (ERDS)	ME-Net
[50]	2023	C4 injury (1)	2 MEA (96)	L/ R/ B reach-	MI	Firing rate	PETH	SVM
[38]	2023	C4 injury (1)	MEA (96)	L/ R/ B/ Re wrist/elbow	MI	LFP	Average	RF
[30]	2024	C4 injury (1)	2 MEA (96)	L/ R/ B cursor control	MA	Firing rate	- -	RNN
[66]	2024	Healthy (100)	EEG (64)	L/ R/ B/ F	ME/ MI	ERDS	FD; relative	SVM/ GB/ kNN/ RF
[49]	2024	Healthy (9)	EEG (32)	L/ R/ B reach-	ME	MRCP	MRCP am- plitudes	LSDA

the emerging trend of bimanual coordination in BCI, necessitating a thorough exploration of its complexities.

III. SYNTHESIS OF RESULTS

This section is organized into six subsections, each exploring variations in studies across different stages of the BCI pipeline. As illustrated in Fig. 3, the general BCI framework consists of two main components: signal acquisition and signal analysis. Participants—whether healthy individuals, patients, or non-human primates—perform specially designed bimanual motor tasks, while their brain signals are recorded using non-invasive (e.g., EEG, MEG, fNIRS, and fMRI) or

TABLE II

STUDIES ON BIMANUAL COORDINATION IN BRAIN-COMPUTER INTERFACES EXCLUDED FROM FINAL REVIEW FOR LACK OF NEUROLOGICAL INSIGHTS OR DISCUSSION ON KEY FEATURES IN DECODING BIMANUAL MOVEMENTS

Ref.	Year	Sensors (n)	Task	Paradigm	Objective
[118]	2011	EEG (5)	B/ F opening- closing (MI); flickering visual stimulus (SSVEP)	MI/ SSVEP	Compared ERDS-based, SSVEP-based and hybrid BCI approaches.
[119]	2014	EEG (14)	L/ R/ B	MI	Used EEG signals to control a car rac- ing game simulator and evaluate the per- formance of different spatial filtering tech- niques.
[120]	2015	EEG (8); tEEG (8); fNIRS (16)	L/ R/ B/ F	MI	Evaluated multimodal BCI using different sensor combinations.
[121]	2015	EEG (14)	L/ R/ B	MI	Utilized BCI to control the altitude of a quadrotor helicopter.
[122]	2016	EEG (64)	L/ R/ B thumb or fifth finger flexion	MI	Proposed a new spatiotemporal and spectral feature extraction method to classify finger flexion MI.
[123]	2017	2 ECoG (4x8)	L/ R/ B/ Re reach-and-press	ME	Proposed a hybrid method to decode move- ment conditions and hand trajectories from NHP neural signals.
[124]	2017	EEG (6)	Flickering visual stimulus	SSVEP	Developd an online BCI system using multi- channel SSVEP to control a dual-arm robot
[125]	2017	EEG (24)	L/ R/ B/ Re	MI	Designed a brain-controlled wheelchair with five steering behaviors using a two-stage control strategy that combines sustained and brief MI.
[126]	2018	EEG (20)	L/ R/ B/ F	MI	Integrated collaborative strategies with BCI to enhance the performance.
[127]	2018	EEG (8)	L/ R/ B/ F	MI	Developed feature extraction methods using ITD and FNN to decode MI.
[128]	2018	EEG (3)	L/ R/ B/ Re closing	ME/ MI	Developed a system to control two robotic hands in real-time using Hjorth parameters and FNN.
[129]	2018	EEG (9)	L/ R/ B/ F grasping	MI	Developed a classification scheme for eight types of MI including multitasking tasks such as L+F, R+F, and B+F.
[130]	2018	2 ECoG (N/A)	L/ R/ B/ Re reach-and-press	ME	Developed a CNN for classifying movement states from NHP cortical signals using spec- trogram inputs.
[131]	2018	MEG (114)	B opening- closing	ME/ MI	Decoded and controlled a humanoid robot performing simultaneous bimanual move- ments using MEG amplitudes in real-time.
[132]	2018	MEG (114)	B opening- closing	ME/ MI	Decoded and controlled a humanoid robot executing simultaneous bimanual move- ments using low-frequency MEG compo- nents in real-time.
[33]	2019	2 ECoG (64)	L/ R/ B reach- and-touch and wrist rotations	MI/ MA	Designd a four-limb neuroprosthetic ex- oskeleton controlled by continuous online epidural ECoG in a tetraplegic patient.
[28]	2019	MEA (164- 171)	L/ R/ B reach- ing	ME	Used LSTM to decode bimanual reaching tasks from single-unit recordings in NHP.

TABLE II

(Continued.) Studies on Bimanual Coordination in Brain-Computer Interfaces Excluded from Final Review for Lack of Neurological Insights or Discussion on Key Features in Decoding Bimanual Movements

Ref.	Year	Sensors (n)	Task	Paradigm	Objective
[133]	2020	EEG (16)	L/ R/ B/ F fist- ing	MI	Used a CNN with spectrograms of CSP- filtered signals as input to decode MI tasks.
[134]	2020	EEG (128)	L/ R/ B/ Re opening-closing	MI	Identified the spatio-temporal aspects of EEG-based continuous neurorobotics where users control a robotic arm and virtual cursor to track target using MI.
[135]	2020	EEG (16)	В	MI	Decoded the termination of motor imagery to enhance closed-loop MI-BCI system.
[27]	2021	EEG (24); fNIRS (24)	B Isometric contraction	ME	Utilized multimodal BCI to decode con- tinuous bimanual grip force through deep learning.
[136]	2022	EEG (16)	B/ F	MI	Proposed a control system for continuous teleoperation of a robotic manipulator using BCI.
[137]	2022	EEG (24)	B multi- direction reaching	ME	Decoded multi-directional bimanual move- ments from EEG potential amplitudes and power sums.
[138]	2022	EEG (64)	L/ R/ B/ F opening-closing	MI	Utilized resting-state EEG alpha rythm to predict subject performance in MI-BCI.
[32]	2022	3 MEA (96); 3 MEA (32)	L/ R/ Re hand open; finger pinch; wrist flexion/ extension	ΜΑ	Developed a collaborative shared control approach to manipulate prosthetic limbs for performing a bimanual self-feeding task in a SCI patient.
[139]	2023	EEG (64)	L/ R/ B/ F/ Re opening-closing	MI	Proposed a novel method called AutoEncoder-Filter Bank Common Spatial Patterns (AE-FBCSP) for decoding imagined movements from EEG signals.
[140]	2023	EEG (29/ 64)	L/ R/ B/ F opening-closing	ME/ MI	Conducted a study on the performance of a novel implementation of the Riemannian geometry decoding algorithm.
[141]	2024	EEG (15)	L/ R/ B rotation (lateral/medial)	ME	Used a CSP-LR pipeline to decode uniman- ual, semi-asymmetric bimanual and symmet- ric bimanual movement.

invasive (e.g., ECoG and MEA) neuroimaging techniques. The acquired signals are processed using various decoding strategies. Decoding may occur simultaneously with signal collection (on-line decoding) or after the complete dataset has been acquired (off-line decoding). Depending on the approach, traditional algorithms, machine learning models, or deep learning techniques are employed to interpret the signal. The decoded outputs can then be utilized to provide real-time feedback—such as visual cues—or to control external devices, including robots and actuators.

A. Study Population

Figure 4a illustrates the distribution of the study population. Thirty-two studies included human participants, with 23 focusing on healthy participants. Notably, this review did not define healthy participants based on demographic factors such as age or sex. Instead, they were defined as individuals without known neurological or psychiatric diseases, without motor deficits, and with normal or corrected-to-normal eyesight. The activation of brain networks and interhemispheric interactions necessary for bimanual movements is influenced by non-pathological factors, including handedness and motor skills [12], [42]. However, interpreting and generalizing these findings poses challenges due to limited reporting of hand dominance in numerous studies. Specifically, while some studies exclusively recruited right-handed participants [23], [25], [26], [30], [31], [34], [35], [41], [43], [44], [45], [46], [47], [48], [49], [50], [51], [52], [53], [54], only studies [55], [56] included both right-handed and left-handed individuals. Furthermore, Riquelme et al. [35] compared BCI control performance between right-handed participants with high bimanual dexterity (pianists) and control participants





Fig. 2. Visualization of eligible articles by publication type (pie chart) and year (bar graph).

(non-pianists). In contrast, Edelman et al. [29] recruited participants with varying BCI experience, including both rightand left-handed individuals.

Several studies recruited patients exhibiting motor deficits due to spinal cord injury (SCI) or spinocerebellar degeneration (SCD). These conditions often result in permanent changes in strength, sensation, and other bodily functions below the affected site. Four studies focused on patients with SCI [30], [31], [38], [50], except for [53], which included both SCI and SCD patients. The study [39] involved a tetraplegic patient but did not specify the type of disease or injury. In addition, the study described in [51] recruited stroke patients and healthy participants, with the healthy group further divided into older and younger subgroups, to evaluate the effects of age and stroke on neuroplasticity during BCI feedback sessions involving muscle stimulation. A study detailed in [55] compared BCI performance between individuals diagnosed with SCI and healthy control participants, while [57] examined EEG features and classification methods in both healthy individuals and patients with disorders of consciousness (DOC), including patients with a minimally conscious state (MCS), and those with unresponsive wakefulness syndrome (UWS). Furthermore, four studies incorporated non-human primates (NHPs) [24], [58], [59], [60].

B. Neuroimaging Techniques

In this review, six neuroimaging techniques were identified and categorized, as shown in Fig. 4b. These included non-invasive methods such as electroencephalography (EEG), magnetoencephalography (MEG), functional near-infrared spectroscopy (fNIRS), and functional magnetic resonance imaging (fMRI). Additionally, invasive techniques, namely electrocorticography (ECoG) and microelectrode array (MEA), were considered. Each method provides distinct insights into brain function [2], [61], [62]. Among the studies analyzed, 26 employed non-invasive techniques, while 10 utilized invasive methods.

1) Non-Invasive Neuroimaging: Non-invasive brain recording involves monitoring and measuring brain activity without surgical intervention or skull penetration. EEG captures electrical signals via scalp electrodes, providing high temporal resolution but limited spatial resolution. MEG detects magnetic fields generated by neuronal activity, offering high temporal and spatial resolution despite cost and accessibility challenges. fNIRS assesses blood oxygenation level-dependent (BOLD) signals using near-infrared light, providing moderate temporal resolution and detailed brain activity mapping by observing BOLD signals, although it sacrifices temporal resolution.

Among studies employing non-invasive neuroimaging techniques, EEG emerged as the most frequently utilized method, featured in 20 studies [23], [25], [34], [35], [41], [43], [45], [46], [47], [48], [49], [51], [52], [54], [55], [57], [63], [64], [65], [66]. Additionally, MEG was employed in three studies [40], [56], [67], and fNIRS was utilized in one study [44].

Moreover, two studies utilized multimodal non-invasive techniques. EEG and fNIRS signals were jointly employed to decode bimanual isometric contraction tasks and evaluated the efficacy of using either or both methods [26]. In a separate investigation [29], fMRI estimated cortical activity sources and improved EEG signal quality by mitigating volume conduction effects.

2) Invasive Neuroimaging: ECoG records cerebral cortex electrical activity by placing electrodes directly on the brain surface, yielding high spatial resolution compared to non-invasive methods such as EEG. In contrast, MEA achieves high spatial resolution by simultaneously recording multiple neurons, often inserted into brain tissue. ECoG, conversely, captures neural signals with high fidelity and is suitable for mapping cortical activity related to specific functions. Meanwhile, MEA enables the study of neuronal populations and synaptic connectivity within the brain.

All invasive studies were conducted on either NHPs or patients. Specifically, three studies employed ECoG in NHP research [24], [59], [60], while one study focused on patients [31]. Additionally, one study utilized MEA in NHP research [28], [58], and five studies involved patients [30], [38], [39], [50], [53].

Studies [30], [39] have highlighted the use of both invasive and non-invasive methods. However, data from noninvasive techniques, such as fMRI and MEG, were exclusively employed for surgical guidance or electrode placement and were not analyzed.

C. Experimental Task and Paradigm

Bimanual coordination tasks exhibit varying difficulty levels based on the degree of hand symmetry. Both discrete



Fig. 3. Generalized BCI pipeline of bimanual motor coordination studies included in this review paper.

and continuous bimanual coordination patterns present challenges when limbs are assigned distinct movement directions, velocities, or amplitudes. In such scenarios, individuals must coordinate their movements across both hands, necessitating precise timing and control. These variations in movement parameters increase task complexity, demanding heightened cognitive and motor control for effective bimanual coordination. This review identified several studies that have incorporated various unimanual and bimanual tasks. These tasks range from simple unimanual left/right-hand (L/R) movements to more complex activities, such as bimanual tapping and tracking.

The experimental tasks employed in these studies encompassed paradigms, such as motor execution (ME), motor imagery (MI), motor attempt (MA), and action observation (AO), as shown in Fig. 4c. Although these paradigms represent different cognitive tasks, the brain activation elicited by each paradigm shares some similarities [68], [69], [70], [71]. Figure 5 presents examples of bimanual motor tasks, with the illustrations adapted from the experimental paradigms of the studies included in this review paper.

1) Motor Execution Paradigm: ME involves overt intentional bodily movements, enabling researchers to study natural brain activity and measure factors such as muscle activation, movement trajectory, and force [72], [73]. Nevertheless, ME paradigms do not apply to patients lacking residual movement.

Sato et al. [44] instructed participants to perform a bimanual ball-grasping task. Kajal et al. [67] utilized a bimanual finger-tapping task with real-time neurofeedback to observe the changes in interhemispheric functional coupling and their impact on motor performance. Additionally, in a study by Ortega et al. [26], noninvasive BCI techniques were employed to decode bimanual grip force. An isometric grip-force tracking task (Fig. 5, *upper right*) was introduced, allowing continuous monitoring and evaluation of force-related brain signals to assess decoding performance.

The reaching task has also received extensive attention. In this task, participants are required to move their hands from one point to another point in space. Zhang et al. [23] decoding reaching direction in one-dimensional space, while Chen et al. [25] focused on reconstructing reaching trajectories, including position and velocity. The researchers in [45] and [48] focused on classifying hand movement directions in a two-dimensional space where the movements are orthogonal. Wang et al. [49] investigated the neural correlates and movement decoding associated with simultaneous and sequential bimanual reaching.

In typical daily scenarios, reaching movements are often combined with other actions to achieve specific goals. Previous research has investigated reach-and-press tasks in NHP studies, where NHPs extended their hands to press buttons [24], [59], [60]. Additionally, Schwarz et al. [41] focused on decoding reach-and-grasp tasks, which involved unimanual lateral or palmar grasp, and bimanual combinations of palmar and lateral grasps or double lateral grasps (Fig. 5, *lower right*).

2) Motor Imagery Paradigm: MI is a cognitive process during which an individual internally rehearses or simulates a given action within their working memory, without movement execution. This practice involves no external motor action or output. The MI paradigm is commonly categorized into two types: kinesthetic MI and visual MI. Kinesthetic MI entails



Fig. 4. (a) Distribution of participants across studies, including healthy individuals (23 studies), patients (6 studies), and non-human primates (4 studies). Three studies recruited both healthy participants and patients. (b) Types of neuroimaging techniques utilized across studies, with most employing non-invasive techniques. (c) Studies employed experimental paradigms, predominantly utilizing motor execution (ME) or motor imagery (MI), followed by motor attempt (MA) and action observation (AO). Additionally, some studies explored multimodal paradigms.

participants mentally simulating muscle activities and movements from a first-person viewpoint. In contrast, visual MI involves imagining movement from a third-person perspective, emphasizing action visualization over muscle sensation [69]. All the studies included in this section utilized kinesthetic MI as their approach.

Two studies utilized unimanual MI tasks. King et al. [34] instructed participants to perform unimanual hand-rotation tasks to investigate changes in brain connectivity and MI performance under various neurofeedback conditions, including bimanual rotation neurofeedback. Additionally, Riquelme et al. [35] employed a task involving unimanual finger drumming with swinging wrists to compare MI performance between pianist and non-pianist groups.

The EEG Motor Movement/Imagery Dataset V1.0.0 (MMIDB) [74], [75], [76] comprises recordings of unimanual,

bimanual, and foot movements performed under both ME and MI conditions. Kim et al. [63] exclusively utilized the MI data from this dataset to evaluate the classification performance of various task combinations. Similarly, studies such as [47], [52], [64], and [65] employed the same set of tasks, whereas [43], [46] extended their analyses to include resting-state (Re) data, though these studies collected their own datasets.

Additionally, Lin et al. [38] and Wan et al. [39] investigated unimanual and bimanual elbow/wrist flexion from neurons, specifically in the left primary motor cortex (Fig. 5, *upper left*). Lai et al. [50] captured neural signals from the patient's left motor cortex during center-out reaching tasks, quantifying the representational interaction between arms by analyzing the tuning parameters of individual neurons. A study by [56] aimed to map and decode bimanual hands and feet MI movements, as well as mental arithmetic (SUB) and silent word generation tasks (WORD). Additionally, participants were instructed to perform MI involving both hand movements as well as the act of holding both hands still in the study [57]. Edelman et al. [29] employed EEG signals with neurofeedback to facilitate participants in executing target-reaching tasks via a virtual cursor and real-time control of a robotic arm.

3) Motor Attempt Paradigm: MA denotes the endeavor to move a paralyzed hand with minimal or no actual movement, typically observed in patients with motor disabilities. This review identified one study that exclusively focused on MA in bimanual BCI. A study in [30] demonstrated the capacity of neural network decoders to enable patients to simultaneously control two computer cursors by attempting bimanual movements.

4) Action Observation Paradigm: AO entails actively observing actions with the intent to imitate. This process can activate mirror neurons, a distinct class that fires during observation and execution of similar motor actions [77], [78]. While none of the reviewed studies specifically addressed AO, studies [51], [53], [58] included AO as part of their multimodal paradigms, as discussed in the subsequent section.

5) Multimodal Paradigm: Several studies have also incorporated more than one paradigm to compare the performance of the proposed algorithms and neural correlates under different experimental conditions. By including multiple paradigms, researchers can assess the influences of variations in task demands, cognitive processes, and neural activity patterns on the effectiveness of their methods.

In their research, Belkacem et al. [40] focused on decoding neuromagnetic activities within the sensorimotor cortex during MI and ME tasks involving symmetric and asymmetric hand movements (Fig. 5, *lower left*). Their objective was to enhance real-time control of humanoid robots using BCI systems. Conversely, Moaveninejad et al. [66] evaluated their proposed classification algorithms using both ME and MI data from MMIDB. Additionally, Vuckovic et al. [54] aimed to classify unimanual and bimanual hand-waving tasks and investigate the specific brain activity associated with each task during MI and ME.

The experiment detailed in [31] directed patients to perform MI or MA without explicit instructions. These actions were subsequently harnessed to control both an exoskeleton



Fig. 5. Bimanual experimental tasks illustrated based on the paradigm explanations of the included studies. Upper left: adapted from [38], [39]; lower left: adapted from [40]; upper right: adapted from [26]; lower right: adapted from [41].



Fig. 6. Input and feature for decoding bimanual motor tasks included review papers. Upper left: ERD/ERS (IEEE image credit: [23]); lower left: neuron firing rate (Nature image credit: [30]); upper right: MRCP (IEEE image credit: [41]); lower right: fNIRS (IOP image credit: [26]).

and its virtual counterpart. The tasks involved bimanual three-dimensional reaching and hand rotation, resulting in an eight-dimensional control space. In contrast, the study described in [55] instructed patients and healthy participants to

execute reach-and-grasp tasks for the MA and ME paradigms, respectively. The researchers then analyzed movement-related cortical potentials (MRCP) and compared decoding performance between the two groups. Kumari et al. [51] recruited

subacute stroke patients, older healthy adults, and younger healthy adults to perform MA or ME combined with AO of wrist extension and flexion. The study aimed to investigate neuroplasticity in bimanual BCI with functional electrical stimulation (FES) feedback. Downey et al. [53] instructed participants to perform MA while observing (AO) a computer executing the same task. The study focused on motor cortical activity associated with unimanual reaching, alternating L/R-hand reach-and-grasp tasks, and bimanual reaching movements.

In their study, Ifft et al. [58] trained NHPs to operate a bimanual BCI without overt movements, utilizing MA. To ensure this, the NHPs' arms were restrained and concealed with an opaque material. This study devised a setup to replicate the real-world challenges faced by paralyzed individuals learning to operate BCIs without overt upper limb movements. The classifiers were trained either through manual control of two joysticks (ME) by the NHPs or by observing avatar arm movements (AO). Additionally, the study investigated neuroplasticity during the NHPs' training with the bimanual BCI.

D. Input and Feature Extraction

Recent advances in deep learning facilitate direct raw data analysis and automatic learning of discriminative representations, reducing reliance on manual feature extraction. EEG studies [23], [25] and the MEA study [30] employed deep learning models without relying on handcrafted feature extraction. In contrast, the EEG-fNIRS study [26] utilized raw fNIRS data (Fig. 6, *lower right*) while incorporating handcrafted feature extraction for EEG signals. In contrast, the EEG study by Abdalsalam et al. [47] utilized an autoregressive model to calculate the frequency spectrum from Laplacian-filtered signals, which was then used to provide visual feedback during MI training.

Nonetheless, many reviewed studies relied on manual feature extraction methods. Moreover, given the diverse aspects of different neuroimaging techniques, the employed feature extraction methods exhibit variation. Brain signals typically encode temporal, spatial, and spectral information relevant to movements. Therefore, feature extraction is crucial for identifying and characterizing these distinct neural aspects.

EEG, MEG, and ECoG studies investigate neural correlates of movements, including movement-related cortical potentials (MRCP) (Fig. 6, *upper right*) and event-related desynchronization/synchronization (ERDS) (Fig. 6, *upper left*). MRCP, referring to low-frequency potentials below 10 Hz, is associated with motor planning and execution processes in the brain [79], [80]. These potentials typically appear approximately 1-2 seconds before voluntary movement onset, enabling the classification of movement intention before its actual execution. In their study, Wang et al. [45] extracted temporal and spectral features from MRCP amplitudes and power sum, respectively. Conversely, studies in [41], [49], and [55] solely utilized MRCP amplitudes feature for their decoding scheme.

ERDS, however, involves time-locked changes in relative power during motor tasks [81]. This analysis typically focuses

on the alpha (7–13 Hz) and beta (13–30 Hz) frequency bands, but it may also encompass other frequencies, such as delta (<4 Hz), theta (4-7 Hz), and gamma (>30 Hz). Spatial features were extracted using variations of the common spatial pattern (CSP) algorithm in several studies [34], [35], [43], [46], [52], [54], [63], [64], while spectral features were derived using the Hilbert transform in the work by Ortega et al. [26]. Moaveninejad et al. [66] extracted temporal and spectral features by estimating fractal dimension and calculating EEG relative power across different bands. Wang et al. [48] described a study where they extracted spectral features from both MRCP and ERDS using wavelet transform. Edelman et al. [29] introduced a co-adaptive BCI to enhance user engagement and improve the spatial resolution of noninvasive neural data. They achieved this through a continuous tracking task that utilized the weighted power density of the EEG alpha band for real-time robotic arm control. Additionally, their training approach aimed to enhance spatial resolution using a novel EEG source imaging technique called frequency-domain electrical source imaging. Kumari et al. [51] employed fast Fourier Transform (FFT) to calculate power in specific frequency bands for providing FES feedback. In contrast, Mohamed et al. [65] proposed leveraging subject-specific regions of interest (ROIs) combined with intrinsic time-scale decomposition (ITD) to improve classification performance. The study in [57] compared features extracted using various methods, including Hjorth parameters, Brainrate, Wackermann features, Hurst exponent, FFT, coherence, Granger causality (GC), partial directed coherence (PDC), directed transfer function (DTF), approximate entropy, Shannon entropy, Bhattacharyya distance, and correlation entropy.

In a study by Belkacem et al. [40], MEG amplitudes were extracted from the event-related magnetic field (ERF). To reduce feature dimensionality, they calculated the mean amplitude of single trials for each MEG sensor around the sensorimotor area. A linear kernel was employed to calculate the group cross-similarity of task-related beta-decrements in ERF, which were then used as input features in the study by Youssofzadeh et al. [56]. In another study by Kajal et al. [67], neurofeedback training utilized the ERF magnitude-squared coherence of sensorimotor rhythm between hemispheres. All ECoG studies employed similar feature extraction methods, specifically wavelet transform, to capture spatial-spectral-temporal features of ECoG signals [24], [31], [59], [60].

Sato et al. [44] introduced the general linear model (GLM) to extract features and mitigate scalp hemodynamic artifacts in fNIRS signals, comparing its performance with several alternative methods Conversely, the majority of MEA studies have employed peri-event time histograms or neuron firing rates (Fig. 6, *lower left*) [30], [39], [50], [53], [58]. However, in a specific MEA study [38], the average power signal of local field potentials was used as a feature.

E. Decoding Algorithm

1) Non-Machine Learning Algorithm: Kajal et al. [67] explored the causal relationship between interhemispheric functional coupling and bimanual performance using neurofeedback. Neurofeedback was determined based on dynamically updated neural coherence, calculated over the preceding trials. In a study by Edelman et al. [29], the instantaneous control signal was computed as the weighted sum of alpha powers from a subset of electrodes. In the study by Kumari et al. [51], tactile feedback was controlled using a simple threshold switch, where the power in the selected frequency band needed to stay below the power threshold for a specific duration. Additionally, Ifft et al. [58] utilized an unscented Kalman filter (UKF) to extract motor commands from brain activity. This approach captures both reaching parameters, such as position and velocity, and their nonlinear relationships with neuronal rates.

2) Machine Learning Algorithm: Various machine learning classifiers have been employed in bimanual BCI research, including variations of the Support Vector Machine (SVM) [39], [40], [44], [45], [46], [50], [52], [55], [56], [57], [64], [66], Linear Discriminant Analysis (LDA) [34], [35], [41], [45], [49], [54], [55], [57], [63], Gaussian Process Classification (GPC) [56], and Relevance Vector Machine (RVM) [64]. Additionally, Random Forest (RF) [38], [55], [66], the Naive Bayes Classifier (NBC) [55], and k-Nearest Neighbors (kNN) [57], [58], [66] have been utilized.

Choi et al. [59] devised a two-stage decoder that combines an effector classifier using LDA and a movement trajectory predictor using Partial Least Squares (PLS) regression. Moaveninejad et al. [66] assessed their feature extraction algorithm by employing SVM, kNN, RF, and a Gradient Boosting classifier (GB) to explore diverse model combinations via soft and hard voting techniques.

In a recent study by Moly et al. [31], novel algorithms were introduced, including Recursive Exponentially Weighted Multisource Linear Modeling (REW-MSLM) and Recursive Exponentially Weighted N-way Partial Least Squares (REW-NPLS). REW-MSLM, an online tensor-based fully adaptive mixture of multilinear expert algorithms, integrates a recursive model parameter identification procedure inspired by the REW-NPLS method [82] and inherits the structure of a Mixture of Linear Modeling (MSLM) [83]. Lindig-Leon and Bougrain [43] employed stepwise regression for classification, calculating CSP and training the model separately for each contralateral movement. In contrast, Abdalsalam et al. [47] used an autoregressive filter to compute the frequency spectrum for controlling cursor movement.

Moreover, Downey et al. [53] employed the Indirect Optimal Linear Estimation (OLE) technique, incorporating ridge regression [84], to create a neural decoder based on an encoding model. This decoder establishes the relationship between neural unit firing rates and unilateral arm velocity, leveraging data from observation trials.

3) Deep Learning Algorithm: Deep learning methods are widely used for decoding neural signal movements. Notably, Convolutional Neural Networks (CNNs) and Recurrent Neural Networks (RNNs) emerge as favored options. CNNs excel at extracting spatial-spectral-temporal features from brain signals, while RNNs capture temporal dependencies and dynamics, facilitating efficient representation learning and classification of mental states or motor intentions [85]. Lim et al. [24] adopted DenseNet [86] to classify NHP arm movements. Meanwhile, Choi et al. [60] extended the DenseNet model to classify three-dimensional ECoG data related to arm movement by incorporating temporal, spectral, and spatial information domains. Similarly, Deo et al. [30] employed Gated Recurrent Unit (GRU), a type of RNN, for movement decoding.

Chen et al. [25] and Zhang et al. [23] combined a well-known CNN architecture, EEGNet, with Long Short-Term Memory (LSTM) for trajectory fitting and decoding, respectively. Additionally, Wang et al. [48] incorporated an attention-based channel-weighting module and CNN to propose ME-NET, which enhances temporal-spectral-spatial feature extraction for movement decoding. In a comprehensive study, Ortega et al. [26] employed CNNs with attention to decoding bimanual force using raw EEG, EEG Hilbert features, and fNIRS. The model was further extended to incorporate diverse architectures, including CNNs with LSTMs, mixture density networks (MDNs), and residual modules with self-attention.

Furthermore, Mohamed et al. [65] employed a simple feedforward neural network (FNN) as their decoding model. In contrast, Tang et al. [64] utilized a spiking neural network (SNN), which closely emulates the structure and function of natural neural networks.

IV. DISCUSSION

A. Toward Practical Bimanual BCI Implementation

Bimanual BCI research aims to enhance movement range and enable complex simultaneous actions, ultimately facilitating patients to perform ADL with greater ease. Consequently, numerous studies have prioritized improving the robustness and real-time performance of the movement decoder.

The key themes in bimanual studies center on achieving high accuracy in movement classification. This pursuit establishes a robust foundation for investigating influential features and models in bimanual decoding and comprehending brain activation during bimanual BCI tasks. However, the limitation of a predefined set of classes may restrict real-world applications. To address this limitation, The decoders providing precise position and velocity trajectories for bimanual movement, and bimanual-controlled cursor trajectory were proposed in studies [25], [30], [58]. Another notable advancement is the continuous decoder developed by Ortega et al. [26] that provides force trajectory information and further enhances the pursuit of a realistic decoder. Moly et al. [31] introduced a highly robust bimanual movement scheme using a Mixture of Expert (MoE) technique. This technique integrates five specialized experts, each responsible for distinct movement tasks: hand translation, hand rotation, and idle movements. These experts collaborate through a gating mechanism akin to a classifier. By applying MoE to this approach, subjects achieve approximately 75% accuracy in controlling virtual avatars and exoskeletons, even in online decoding.

Real-time functionality is pivotal in practical implementation. Belkecem et al. [40] effectively controlled a humanoid robot to execute contralateral and ipsilateral bimanual tasks, achieving average real-time accuracies of 61.25% and 63.5%, respectively, which is deemed acceptable for online decoding. However, when aiming to broaden movement constraints by developing real-time continuous decoders, the performances of these decoders were slightly lower due to the wider movement range. For instance, in studies [31], [53], [58], real-time decoders for virtual avatar and exoskeleton control achieved modest ground-truth correlations. Notably, Handelman et al. [32] introduced an online discrete decoder that enables continuous control via a shared control strategy, mapping subject gestures to specific robot-arm movements. This approach enabled participants to perform real-time selffeeding tasks despite the relatively lower accuracy of the decoder. Furthermore, Deo et al. [30] applied the cursor trajectory decoder in a real-time context. Even though it effectively decoded alternating bimanual task, it struggled to achieve the same performance in simultaneous tasks due to the suppressed left-hand representation in bimanual simultaneous movement. Nonetheless, the efficacy of real-time implementation may vary due to factors such as intersubject variability, fatigue, and emotional state [87].

The main challenge is that most studies focus on either discrete actions or continuous limb translation, which limits generalizability across datasets and populations. This presents an opportunity for future research to integrate both movement types, enabling more naturalistic control in bimanual BCI systems. While collecting large datasets for training decoders is ideal, it is often time-consuming and costly. Adopting transfer learning to generalize motor task variability and address both intrasubject and intersubject non-stationarity may offer a more efficient solution, requiring minimal data collection and model retraining [88], [89]. This approach streamlines development and improves generalization across diverse populations and experimental conditions.

B. Motor Function Restoration in Paralyzed Patient

BCI therapy for post-stroke upper limb paralysis is moderately recommended, with medium evidence support, according to the 2021 Japanese Guidelines for the Management of Stroke by The Japan Stroke Society [90]. This recognition highlights BCI-based therapies' increasing interest and potential in neurorehabilitation. However, most bimanual BCI studies have exclusively focused on SCI patients. Nevertheless, there exists a substantial body of research employing BCI technology for various other conditions that result in motor function impairment, including stroke, cerebral palsy, Parkinson's disease, and multiple sclerosis. Although conclusive evidence of neuroplasticity during bimanual training is lacking, numerous studies have aimed to expand the movement range of paralyzed patients by developing decoding techniques for bimanual tasks. The robust classification accuracies observed in predefined bimanual tasks classification validate the feasibility of simultaneous movement decoding in patients with SCIs [38], [39], [50], [55]. Additionally, advanced continuous bimanual decoders enabled SCI patients to perform target-reaching tasks by controlling virtual avatars, exoskeletons, and cursors, enhancing the potential for motor function restoration [30], [53].

The delta-alpha ratio (DAR) and brain symmetry index (BSI) are commonly used metrics for assessing recovery levels after neurorehabilitation, with lower values indicating stroke recovery [91], [92], [93]. In the study by [51], these indices were used to evaluate the short-term priming effects of unimanual and bimanual BCI-FES feedback across three groups: subacute stroke patients, older healthy adults, and younger healthy adults. However, significant changes in DAR and BSI were observed only in the healthy groups. The authors suggest that injured brains may require longer sessions to induce short-term changes or that gradual improvements could emerge through cumulative effects in long-term studies. They further recommend bimanual BCI-FES as a viable alternative to unimanual BCI-FES.

The inclusion of NHPs, patients, and non-disabled participants in existing bimanual BCI studies presents challenges for transferring domain knowledge across participant categories. Comparing brain activation across patients with different diseases is inherently biased, and even within the same disease, the neurological injury level should be considered. Furthermore, bimanual BCI studies cited in this review likely conform to distinct injury level standards. Nevertheless, interhemispheric interaction, critical for neural activation during bimanual movement, can be influenced by age and sex, as demonstrated by Takeuchi et al. [12]. Similarly, Pfurtscheller et al. [94] observe that ERD strength and lateralization in SCI patients may vary with the recovery course. Chronic complete SCI patients tend to exhibit weaker ERD compared to subacute incomplete SCI patients, leading to inferior BCI performance.

Hence, conducting experiments comparing diverse subject types is essential to explore the transferability of domain knowledge. Currently, only two studies investigate the difference between healthy participants and patients during bimanual activation [51], [55], while another examines an SCI patient and an SCD patient [53]. The scarcity of research in this area invites further exploration, potentially yielding valuable insights into bimanual BCI across various populations.

C. Bimanual Training on BCI Performance and Neuroplasticity

Bimanual motor training has been demonstrated to modulate interhemispheric interactions, promoting symmetrical brain function and enhancing the efficiency of connections between both hemispheres [12]. This facilitates temporally and spatially coordinated movement of both hands. This may be attributed to intermanual transfer, where motor learning in one hand improves performance of the other hand. Amemiya et al. [95] showed that intermanual transfer can occur during both ME and MI tasks. Moreover, both bimanual and unimanual movements benefit from bimanual training, and vice versa [10]. However, the transfer of learning may be limited depending on the type of movement [96].

Riquelme et al. [35] conducted a groundbreaking comparative study on pianists who attained advanced bimanual dexterity through rigorous training, alongside a control group, during MI tasks. They discovered a statistically significant higher BCI performance and lower activation of the bilateral motor cortex in the pianist group during MI. Using brain connectivity analysis, King et al. [34] detected mutual information transfer between prefrontal areas, signifying improved motor preparation and bimanual coordination during bimanual rotational feedback. Additionally, parietal-to-bilateral frontal cortex communication implies that feedback conveys essential spatial location information, critical for motor planning. Study in [31] observed lower bimanual BCI performance in virtual avatar control compared to exoskeleton control, which may be attributed to differences in feedback and perception. Exoskeleton control provides more enriched and realistic feedback, potentially enhancing BCI performance.

The study conducted by Ifft et al. [58] observed performance enhancements in NHPs during real-time BCI control of bimanual movements. Initially, heightened correlated neural activity was observed, likely attributed to the learning process. However, after this initial phase, NHPs consistently exhibited a reduction in correlated neural activity across various cortical areas in both hemispheres as they acquired independent control over both arms. Kajal et al. [67] investigated the causal link between interhemispheric functional coupling and bimanual performance using an out-of-phase bimanual finger-tapping task with neurofeedback. Their findings suggest that stronger functional coupling adversely affects bimanual performance.

Additionally, Edelman et al. [29] compared participant performance in two distinct tasks: discrete center-out cursor control (DT) and continuous pursuit task (CP), both guided by EEG signals during MI. Their findings revealed that CP training outperformed DT, with participants demonstrating superior neural control over both cursor and robotic devices for discrete and continuous target-tracking tasks. Furthermore, higher engagement was demonstrated during training sessions. Moreover, the study investigated the impact of source and sensor neurofeedback on CP training, revealing nearly identical learning effects for both feedback types.

Studies have shown that bimanual BCIs can induce neuroplasticity, but the extent of these changes varies widely among individuals due to factors such as sex, brain disease, and age [12], [97], which poses challenges for achieving consistent neurorehabilitation outcomes. Additionally, different types of activity-based rehabilitation can result in varying levels of brain changes. Personalizing rehabilitation strategies may therefore lead to better outcomes. The brain criticality hypothesis suggests that optimal brain function occurs at or near a critical point between order and disorder, with deviations potentially causing dysfunction [98]. A recent study [99] found that lower brain criticality during resting-state EEG was linked to improved BCI performance in a locked-in patient, while Rocha et al. [100] proposed that personalized brain dynamic models poised near criticality could predict stroke recovery. Future research could potentially employ brain criticality as a metric to tailor bimanual BCI neurorehabilitation, potentially optimizing patient outcomes.

D. Unilateral Cortex Encodes Unimanual and Bimanual Movements

In this review, multiple studies have explored the decoding and neural representation of bimanual movements in the unilateral motor cortex [30], [38], [39], [50], [53], [58]. A consistent finding across these investigations leads to a general inference: neurons within the unilateral motor cortex independently encode both contralateral and ipsilateral movements. Additionally, a significant correlation exists in the representation of movement direction between contralateral and ipsilateral actions. Moreover, their representations exhibit nonlinear changes between unimanual and bimanual movements. The encoding patterns for bimanual movements show a stronger correlation with contralateral movements than ipsilateral movements. Most studies suggest that brain activation patterns during bimanual movements are not simply a superposition of unimanual movements. The observed changes in representations may stem from variations in tuning parameters, such as modulation depths and directional preferences. However, this review found a contradictory finding by Linding et al. [43], who demonstrated effective performance using a stepwise classification strategy, showing that bimanual movements generate similar activity in each brain hemisphere to that produced by simple unimanual motor imagery on the contralateral side.

Notably, Downey et al. [53] observed a correlation in grasping representations between both hands while also identifying distinct representations for hand and arm movements. Previous research [39], [50] identified distinct neural representations for wrist and elbow movements, as well as movement directions. Furthermore, the investigation of unilateral local field potentials revealed distinct representations for bilateral MI, based on average energy across the full array and single-channel power levels [38]. Moreover, considering variations in representations across unimanual, bimanual, and different limb movements within the unilateral motor cortex, it becomes feasible to decode movement types and directions.

Although these studies confirm the involvement of the unilateral cortex in encoding both unimanual and bimanual movements, the underlying mechanisms remain to be fully understood. In unimanual tasks, the ipsilateral hemisphere encodes information distinct from that of the contralateral hemisphere, suggesting its role in both the planning and execution of movements [101]. Potential explanations include balancing interhemispheric inhibition, maintaining an efferent copy of the ipsilateral limb to facilitate bimanual control, and driving proximal muscle activity for posture stabilization. While this remains a topic of debate, investigating the role of both hemispheres in executing unimanual and bimanual movements-especially under task- and time-specific conditions-could significantly enhance the understanding of human motor control mechanisms. This knowledge carries important implications for developing novel rehabilitation strategies for individuals with motor impairments.

E. Important Features for Decoding Bimanual Movements

Studies employing MRCP signals have revealed significant findings. Specifically, they demonstrate a larger negative peak and lateralization of MRCP around movement onset in bimanual movements compared to unimanual movements [23], [45], [48]. Furthermore, a correlation exists between the negative

offset maximum of MRCP and torque level, with higher torque tasks exhibiting greater negative offset maximums than lower torque tasks [102]. For example, Wang et al. [45] argued that upward hand movement against gravity yielded the largest negative amplitudes. Kirchhoff et al. [55] also observed a stronger MRCP rebound during bimanual tasks compared to unimanual tasks. Additionally, they identified an optimal decoding time window approximately 0.75 seconds before movement onset. Furthermore, Schwarz et al. [41] reported slower response times for bimanual movements, potentially indicating the additional planning and cognitive effort required for coordinated bimanual tasks. Wang et al. [49] demonstrated that simultaneous bimanual movements exhibited symmetric activations in both hemispheres, whereas sequential bimanual movements showed asymmetric activations with predominant contralateral effects.

The ERDS studies reveal key brain regions for decoding, including the supplementary motor area (SMA), primary somatosensory cortex (S1), posterior parietal cortex (PPC), and primary motor cortex (M1). Additionally, frequency bands such as delta, alpha, beta, and gamma were associated with decoding motor tasks. Studies [44], [46], [47], [52], [54], [63], [64], [65], [66] observed that during bimanual tasks, both hemispheres are actively engaged, with ERD showing a broader distribution over the non-dominant hemisphere, particularly in the parietal cortex, compared to unimanual tasks. Belkacem et al. [40] detected distinct time and frequency distributions in the SMA across bilateral hemispheres for diverse bimanual tasks. Additionally, bimanual movements were associated with bilateral beta ERD in the SMA, precentral/primary motor cortex, as well as in the anterior cingulate gyri and prefrontal regions areas and the frontal pole [56].

The study in [57] compared various feature extraction and classification methods in healthy participants and patients with DOC performing both hands MI. The results showed that coherence features yielded the best classification results in healthy participants, although unexpectedly, coherence was predominantly observed in the frontal regions. In contrast, none of the extracted features showed significant results in DOC patients, possibly due to the degree of disability and neuroplasticity variations among the patients.

Moreover, NHP studies [24], [59], [60] employing class activation mapping, a branch of xAI, have unveiled significant weights in the ipsilateral motor and somatosensory cortex regions. Simultaneously, gamma-band power activation has been observed in contralateral areas during unimanual movement, suggesting that brain signals acquired from the motor cortex encode information about both contralateral and ipsilateral movements. These results corroborate research demonstrating the decodability of both unimanual and bimanual movements from unilateral neural signals [30], [38], [39], [50], [53], [58]. Furthermore, the hand-movement classification system leveraged critical temporal cues during movement onset and offset, emphasizing its accuracy in bimanual movement classification. Additionally, they highlighted the substantial role of alpha and beta bands in decoding, signifying their importance in hand or finger movement planning and execution. Ortega et al. [26] reported a distinct frontal ipsilateral delta band pattern for the left hand, contrasting with a more bilateral pattern observed for the right hand. This discrepancy likely arises from the dominant cortical representation of each hand. Additionally, they noted contralateralization of the alpha band during bimanual tasks, accompanied by a bilateral frontal-parietal pattern, suggesting involvement of a fine motor control network. Conversely, Kirchhoff et al. [55] observed that feature importance for decoding predominately centered around the motor cortex in healthy participants, but in patients, it shifted to the frontal cortex.

Despite successful decoding of bimanual tasks, cortical activation patterns for different bimanual tasks lack strong spatial distinctiveness, particularly when using non-invasive neuroimaging techniques [41], [54]. Consequently, a significant challenge remains in expanding the DOF or the number of tasks that a bimanual BCI can decode. Furthermore, only a limited number of studies in the review employed multimodal neuroimaging techniques. Adopting such an approach could be advantageous, as multimodal neuroimaging provides a more comprehensive assessment of the integrated neural mechanisms underlying complex processes. This methodology allows for more robust comparisons, cross-validation across modalities, and can significantly enhance the overall understanding of brain function in BCI applications.

F. Future Prospects in Bimanual Coordination in BCI

While the reviewed papers offer diverse research directions, their limitations highlight opportunities for future development. The focus of bimanual BCI studies in this review has primarily been on decoding motor intentions during the execution of MI or ME, and translating these intentions into real movements by controlling peripheral devices. However, achieving precise and real-time control requires a deeper understanding of motor planning-the preparatory stage preceding movements. Studies by Hanakawa et al. [103], [104] have shown differences in the distributed motor network during motor planning, MI, and ME. Although brain regions involved in MI are similar to those in motor planning, discerning these distinctions is crucial for refining BCI control algorithms. A comprehensive grasp of motor planning could enable controllers to discern between preparation signals and actual execution signals, enhancing the system's responsiveness and accuracy. Another promising development prospect is implementing a system capable of halting execution upon user request. Bhattacharyya et al. [105] proposed a robot arm's online control scheme that utilizes the error-related potential signal to stop the movement of individual links. Integrating these ideas into the bimanual BCI system can pursue a more realistic movement.

Moreover, the application of BCIs in the spinal cord is an intriguing research topic in the BCI-based neurorehabilitation field. Several studies [106], [107], [108], [109] have demonstrated the potential of restoring communication pathways in the injured spinal cord. Currently, only one study found the potential of utilizing spinal fMRI and electromyography signals from healthy patients to decode bimanual tasks [110]. While this study may not explicitly focus on motor function restoration, its investigation and analysis contribute to a deeper understanding of spinal cord mechanisms related to bimanual

motor control. Exploring the integration of bimanual BCIs into spinal cord applications presents an intriguing avenue for future research.

The primary goal of most bimanual BCI applications is to restore motor function in patients with deficits, enabling them to regain natural movements through brain-controlled devices. However, it is crucial to explore whether these technologies can also augment the human body, enhancing task performance and enabling multitasking. One promising application is the development of supernumerary limbs, which aim to expand motor capabilities by adding extra limbs [111], [112]. While this offers exciting prospects for BCI, the application is relatively new and requires further research. Specifically, it is important to address the neural allocation problem, which examines whether the brain can adapt to effectively provide control signals to additional limbs without impairing the function of existing ones [113], [114], [115]. Furthermore, the extra limb must provide sensory feedback that complements natural information and creates a sense of realistic embodiment [116], [117].

V. CONCLUSION

This review investigated recent trends in bimanual coordination within BCI research. It comprehensively examined participant demographics, neuroimaging methods, experimental tasks and paradigms, and feature extraction and decoding algorithms. Recommendations to enhance bimanual BCI usability and support motor function rehabilitation in individuals with motor deficits are suggested. The review synthesized key neurological findings on bimanual coordination and proposed future research directions. Leveraging advancements in neurological understanding and BCI technology, it anticipates progress in motor BCIs towards more natural and practical applications. Thus, this article serves as a starting point for designing bimanual motor coordination studies in BCI, addressing both engineering and neurological aspects by summarizing recent trends.

REFERENCES

- U. Chaudhary, N. Birbaumer, and A. Ramos-Murguialday, "Braincomputer interfaces for communication and rehabilitation," *Nature Rev. Neurol.*, vol. 12, no. 9, pp. 513–525, 2016.
- [2] S. N. Abdulkader, A. Atia, and M.-S.-M. Mostafa, "Brain computer interfacing: Applications and challenges," *Egyptian Informat. J.*, vol. 16, no. 2, pp. 213–230, Jul. 2015.
- [3] J. Wang, L. Bi, and W. Fei, "EEG-based motor BCIs for upper limb movement: Current techniques and future insights," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 31, pp. 4413–4427, 2023.
- [4] X. Yong and C. Menon, "EEG classification of different imaginary movements within the same limb," *PLoS ONE*, vol. 10, no. 4, Apr. 2015, Art. no. e0121896.
- [5] D. Achanccaray and M. Hayashibe, "Decoding hand motor imagery tasks within the same limb from EEG signals using deep learning," *IEEE Trans. Med. Robot. Bionics*, vol. 2, no. 4, pp. 692–699, Nov. 2020.
- [6] C. Phunruangsakao, D. Achanccaray, S.-I. Izumi, and M. Hayashibe, "Multibranch convolutional neural network with contrastive representation learning for decoding same limb motor imagery tasks," *Frontiers Human Neurosci.*, vol. 16, Dec. 2022, Art. no. 1032724.
- [7] R. Alazrai, H. Alwanni, and M. I. Daoud, "EEG-based BCI system for decoding finger movements within the same hand," *Neurosci. Lett.*, vol. 698, pp. 113–120, Apr. 2019.
- [8] J. Pereira, A. I. Sburlea, and G. R. Müller-Putz, "EEG patterns of selfpaced movement imaginations towards externally-cued and internallyselected targets," *Sci. Rep.*, vol. 8, no. 1, p. 13394, Sep. 2018.

- [9] A. Schwarz, P. Ofner, J. Pereira, A. I. Sburlea, and G. R. Müller-Putz, "Decoding natural reach-and-grasp actions from human EEG," *J. Neural Eng.*, vol. 15, no. 1, Feb. 2018, Art. no. 016005.
- [10] K. Schulze, E. Lüders, and L. Jäncke, "Intermanual transfer in a simple motor task," *Cortex*, vol. 38, no. 5, pp. 805–815, Jan. 2002.
- [11] M. Trlep, M. Mihelj, and M. Munih, "Skill transfer from symmetric and asymmetric bimanual training using a robotic system to single limb performance," *J. Neuroeng. Rehabil.*, vol. 9, no. 1, pp. 1–14, Dec. 2012.
- [12] N. Takeuchi, Y. Oouchida, and S.-I. Izumi, "Motor control and neural plasticity through interhemispheric interactions," *Neural Plasticity*, vol. 2012, pp. 1–13, May 2012.
- [13] C. Phunruangsakao, J. Budsuren, and M. Hayashibe, "Modulation of alpha band brain connectivity during motor imagery via bimanual motor training," in *Proc. 12th Int. Winter Conf. Brain–Comput. Interface (BCI)*, Feb. 2024, pp. 1–6.
- [14] A. Wolf, R. Scheiderer, N. Napolitan, C. Belden, L. Shaub, and M. Whitford, "Efficacy and task structure of bimanual training post stroke: A systematic review," *Topics Stroke Rehabil.*, vol. 21, no. 3, pp. 181–196, May 2014.
- [15] V. A.-Q. Dong, I. H.-H. Tung, H. W.-Y. Siu, and K. N.-K. Fong, "Studies comparing the efficacy of constraint-induced movement therapy and bimanual training in children with unilateral cerebral palsy: A systematic review," *Develop. Neurorehabilitation*, vol. 16, no. 2, pp. 133–143, Apr. 2013.
- [16] M. Tervahauta, G. Girolami, and G. Øberg, "Efficacy of constraintinduced movement therapy compared with bimanual intensive training in children with unilateral cerebral palsy: A systematic review," *Clin. Rehabil.*, vol. 31, no. 11, pp. 1445–1456, Nov. 2017.
- [17] Y.-C. Hung, L. Casertano, A. Hillman, and A. M. Gordon, "The effect of intensive bimanual training on coordination of the hands in children with congenital hemiplegia," *Res. Develop. Disabilities*, vol. 32, no. 6, pp. 2724–2731, Nov. 2011.
- [18] A. E. Q. van Delden, P. J. Beek, M. Roerdink, G. Kwakkel, and C. E. Peper, "Unilateral and bilateral upper-limb training interventions after stroke have similar effects on bimanual coupling strength," *Neurorehabilitation Neural Repair*, vol. 29, no. 3, pp. 255–267, Mar. 2015.
- [19] A. E. Q. van Delden, C. E. Peper, G. Kwakkel, and P. J. Beek, "A systematic review of bilateral upper limb training devices for poststroke rehabilitation," *Stroke Res. Treat.*, vol. 2012, pp. 1–17, May 2012.
- [20] A. S. Merians, E. Tunik, and S. V. Adamovich, "Virtual reality to maximize function for hand and arm rehabilitation: Exploration of neural mechanisms," *Stud. Health Technol. Inf.*, vol. 145, p. 109, Jan. 2009.
- [21] P. C. Desrochers, A. T. Brunfeldt, and F. A. Kagerer, "Neurophysiological correlates of adaptation and interference during asymmetrical bimanual movements," *Neuroscience*, vol. 432, pp. 30–43, Apr. 2020.
- [22] L. Holper, M. Biallas, and M. Wolf, "Task complexity relates to activation of cortical motor areas during uni- and bimanual performance: A functional NIRS study," *NeuroImage*, vol. 46, no. 4, pp. 1105–1113, Jul. 2009.
- [23] M. Zhang et al., "Decoding coordinated directions of bimanual movements from EEG signals," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 31, pp. 248–259, 2023.
- [24] S. Lim, D. P. Jang, and H. Choi, "Deep learning classification features visualization for arm movement brain-computer interface," in *Proc. 8th Int. Winter Conf. Brain-Comput. Interface (BCI)*, Feb. 2020, pp. 1–3.
- [25] Y.-F. Chen et al., "Continuous bimanual trajectory decoding of coordinated movement from EEG signals," *IEEE J. Biomed. Health Inf.*, vol. 26, no. 12, pp. 6012–6023, Dec. 2022.
- [26] P. Ortega and A. A. Faisal, "Deep learning multimodal fNIRS and EEG signals for bimanual grip force decoding," *J. Neural Eng.*, vol. 18, no. 4, Aug. 2021, Art. no. 0460e6.
- [27] P. Ortega, T. Zhao, and A. Faisal, "Deep real-time decoding of bimanual grip force from EEG & fNIRS," in *Proc. 10th Int. IEEE/EMBS Conf. Neural Eng. (NER)*, May 2021, pp. 714–717.
- [28] P.-H. Tseng, N. A. Urpi, M. Lebedev, and M. Nicolelis, "Decoding movements from cortical ensemble activity using a long shortterm memory recurrent network," *Neural Comput.*, vol. 31, no. 6, pp. 1085–1113, Jun. 2019.
- [29] B. J. Edelman et al., "Noninvasive neuroimaging enhances continuous neural tracking for robotic device control," *Sci. Robot.*, vol. 4, no. 31, p. 6844, Jun. 2019.
- [30] D. R. Deo, F. R. Willett, D. T. Avansino, L. R. Hochberg, J. M. Henderson, and K. V. Shenoy, "Brain control of bimanual movement enabled by recurrent neural networks," *Sci. Rep.*, vol. 14, no. 1, p. 1598, Jan. 2024.

283

- [31] A. Moly et al., "An adaptive closed-loop ECoG decoder for long-term and stable bimanual control of an exoskeleton by a tetraplegic," *J. Neural Eng.*, vol. 19, no. 2, Apr. 2022, Art. no. 026021.
- [32] D. A. Handelman et al., "Shared control of bimanual robotic limbs with a brain-machine interface for self-feeding," *Frontiers Neurorobot.*, vol. 16, Jun. 2022, Art. no. 918001.
- [33] A. L. Benabid et al., "An exoskeleton controlled by an epidural wireless brain-machine interface in a tetraplegic patient: A proof-of-concept demonstration," *Lancet Neurol.*, vol. 18, pp. 1112–1122, Dec. 2019.
- [34] J.-T. King et al., "Brain connectivity changes during bimanual and rotated motor imagery," *IEEE J. Transl. Eng. Health Med.*, vol. 10, pp. 1–8, 2022.
- [35] J.-V. Riquelme-Ros, G. Rodríguez-Bermúdez, I. Rodríguez-Rodríguez, J.-V. Rodríguez, and J.-M. Molina-García-Pardo, "On the better performance of pianists with motor imagery-based brain-computer interface systems," *Sensors*, vol. 20, no. 16, pp. 44–52, Aug. 2020.
- [36] M. J. Page et al., "The PRISMA 2020 statement: An updated guideline for reporting systematic reviews," *Int. J. Surg.*, vol. 88, Mar. 2021, Art. no. 105906.
- [37] H. M. Sisti, A. Beebe, M. Bishop, and E. Gabrielsson, "A brief review of motor imagery and bimanual coordination," *Frontiers Hum. Neurosci.*, vol. 16, Nov. 2022, Art. no. 1037410.
- [38] J. Lin et al., "Representation and decoding of bilateral arm motor imagery using unilateral cerebral LFP signals," *Frontiers Hum. Neurosci.*, vol. 17, Jun. 2023, Art. no. 1168017.
- [39] Z. Wan, D. Lai, F. Ren, W. Chen, and K. Xu, "The primary motor cortex represents unilateral and bilateral movements of elbows and wrists— A pilot study," in *Proc. 7th Int. Conf. Biomed. Signal Image Process.* (*ICBIP*), Aug. 2022, pp. 127–131.
- [40] A. N. Belkacem, S. Nishio, T. Suzuki, H. Ishiguro, and M. Hirata, "Neuromagnetic decoding of simultaneous bilateral hand movements for multidimensional brain-machine interfaces," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 26, no. 6, pp. 1301–1310, Jun. 2018.
- [41] A. Schwarz, J. Pereira, R. Kobler, and G. R. Müller-Putz, "Unimanual and bimanual reach-and-grasp actions can be decoded from human EEG," *IEEE Trans. Biomed. Eng.*, vol. 67, no. 6, pp. 1684–1695, Jun. 2020.
- [42] R. R. Walsh, S. L. Small, E. E. Chen, and A. Solodkin, "Network activation during bimanual movements in humans," *NeuroImage*, vol. 43, no. 3, pp. 540–553, Nov. 2008.
- [43] C. Lindig-Leon and L. Bougrain, "A multi-label classification method for detection of combined motor imageries," in *Proc. IEEE Int. Conf. Syst., Man, Cybern.*, Oct. 2015, pp. 3128–3133.
- [44] T. Sato, K. Sugai, I. Nambu, and Y. Wada, "Classification of functional near-infrared spectroscopy signals applying reduction of scalp hemodynamic artifact," in *Proc. UKACC Int. Conf. Control (CONTROL)*, Jul. 2014, pp. 708–713.
- [45] J. Wang, L. Bi, W. Fei, and C. Guan, "Decoding single-hand and bothhand movement directions from noninvasive neural signals," *IEEE Trans. Biomed. Eng.*, vol. 68, no. 6, pp. 1932–1940, Jun. 2021.
- [46] W. Yi, S. Qiu, H. Qi, L. Zhang, B. Wan, and D. Ming, "EEG feature comparison and classification of simple and compound limb motor imagery," *J. Neuroeng. Rehabil.*, vol. 10, no. 1, pp. 1–12, Dec. 2013.
- [47] E. Abdalsalam, M. Z. Yusoff, A. Malik, N. S. Kamel, and D. Mahmoud, "Modulation of sensorimotor rhythms for brain-computer interface using motor imagery with online feedback," *Signal, Image Video Process.*, vol. 12, no. 3, pp. 557–564, Mar. 2018.
- [48] J. Wang, L. Bi, A. G. Feleke, and W. Fei, "MRCPs-and-ERS/Doscillations-driven deep learning models for decoding unimanual and bimanual movements," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 31, pp. 1384–1393, 2023.
- [49] J. Wang et al., "Neural correlate and movement decoding of simultaneous-and-sequential bimanual movements using EEG signals," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 32, pp. 2087–2095, 2024.
- [50] D. Lai et al., "Neuronal representation of bimanual arm motor imagery in the motor cortex of a tetraplegia human, a pilot study," *Frontiers Neurosci.*, vol. 17, Mar. 2023, Art. no. 1133928.
- [51] R. Kumari et al., "Short term priming effect of brain-actuated muscle stimulation using bimanual movements in stroke," *Clin. Neurophysiol.*, vol. 138, pp. 108–121, Jun. 2022.
- [52] H. Wang, T. Li, A. Bezerianos, H. Huang, Y. He, and P. Chen, "The control of a virtual automatic car based on multiple patterns of motor imagery BCI," *Med. Biol. Eng. Comput.*, vol. 57, no. 1, pp. 299–309, Jan. 2019.

- [53] J. E. Downey et al., "The motor cortex has independent representations for ipsilateral and contralateral arm movements but correlated representations for grasping," *Cerebral Cortex*, vol. 30, no. 10, pp. 5400–5409, Sep. 2020.
- [54] A. Vuckovic, S. Pangaro, and P. Finda, "Unimanual versus bimanual motor imagery classifiers for assistive and rehabilitative brain computer interfaces," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 26, no. 12, pp. 2407–2415, Dec. 2018.
- [55] M. Kirchhoff, S. Evers, M. Wolf, R. Rupp, and A. Schwarz, "Decoding reach and attempted grasp actions from EEG of persons with spinal cord injury," in *Proc. IEEE Int. Conf. Syst., Man, Cybern. (SMC)*, Oct. 2022, pp. 1624–1629.
- [56] V. Youssofzadeh et al., "Mapping and decoding cortical engagement during motor imagery, mental arithmetic, and silent word generation using MEG," *Hum. Brain Mapping*, vol. 44, no. 8, pp. 3324–3342, Mar. 2023.
- [57] Y. Höller et al., "Comparison of EEG-features and classification methods for motor imagery in patients with disorders of consciousness," *PLoS ONE*, vol. 8, no. 11, Nov. 2013, Art. no. e80479.
- [58] P. J. Ifft, S. Shokur, Z. Li, M. A. Lebedev, and M. A. L. Nicolelis, "A brain-machine interface enables bimanual arm movements in monkeys," *Sci. Transl. Med.*, vol. 5, no. 210, Nov. 2013, Art. no. 210ra154.
- [59] H. Choi et al., "Improved prediction of bimanual movements by a two-staged (effector-then-trajectory) decoder with epidural ECoG in nonhuman primates," *J. Neural Eng.*, vol. 15, no. 1, Feb. 2018, Art. no. 016011.
- [60] H. Choi, S. Lim, K. Min, K.-H. Ahn, K.-M. Lee, and D. P. Jang, "Nonhuman primate epidural ECoG analysis using explainable deep learning technology," J. Neural Eng., vol. 18, no. 6, Dec. 2021, Art. no. 066022.
- [61] G. Xue, C. Chen, Z.-L. Lu, and Q. Dong, "Brain imaging techniques and their applications in decision-making research," *Acta Psychologica Sinica*, vol. 42, no. 1, pp. 120–137, Feb. 2010.
- [62] S. Saha et al., "Progress in brain computer interface: Challenges and opportunities," *Frontiers Syst. Neurosci.*, vol. 15, Feb. 2021, Art. no. 578875.
- [63] H. S. Kim, M. H. Chang, H. J. Lee, and K. S. Park, "A comparison of classification performance among the various combinations of motor imagery tasks for brain–computer interface," in *Proc. 6th Int. IEEE/EMBS Conf. Neural Eng. (NER)*, Nov. 2013, pp. 435–438.
- [64] C. Tang, L. Xu, P. Chen, Y. He, A. Bezerianos, and H. Wang, "A novel multiple motor imagery experimental paradigm design and neural decoding," in *Proc. Chin. Autom. Congr. (CAC)*, Nov. 2020, pp. 4024–4028.
- [65] E. A. Mohamed, I. K. Adam, and M. Z. Yusoff, "Effect of subjectspecific region of interest on motor imagery brain-computer interface," *Appl. Sci.*, vol. 13, no. 11, p. 6364, May 2023.
- [66] S. Moaveninejad et al., "Fractal dimension as a discriminative feature for high accuracy classification in motor imagery EEG-based braincomputer interface," *Comput. Methods Programs Biomed.*, vol. 244, Feb. 2024, Art. no. 107944.
- [67] D. S. Kajal et al., "Learned control of inter-hemispheric connectivity: Effects on bimanual motor performance," *Human Brain Mapping*, vol. 38, no. 9, pp. 4353–4369, Sep. 2017.
- [68] Y. J. Yang, E. J. Jeon, J. S. Kim, and C. K. Chung, "Characterization of kinesthetic motor imagery compared with visual motor imageries," *Sci. Rep.*, vol. 11, no. 1, p. 3751, Feb. 2021.
- [69] C. Neuper, R. Scherer, M. Reiner, and G. Pfurtscheller, "Imagery of motor actions: Differential effects of kinesthetic and visual-motor mode of imagery in single-trial EEG," *Cognit. Brain Res.*, vol. 25, no. 3, pp. 668–677, Dec. 2005.
- [70] T. Mulder, "Motor imagery and action observation: Cognitive tools for rehabilitation," J. Neural Transmiss., vol. 114, no. 10, pp. 1265–1278, Oct. 2007.
- [71] S. Chen, X. Shu, H. Wang, L. Ding, J. Fu, and J. Jia, "The differences between motor attempt and motor imagery in brain–computer interface accuracy and event-related desynchronization of patients with hemiplegia," *Frontiers Neurorobotics*, vol. 15, Nov. 2021, Art. no. 706630.
- [72] L. Xu et al., "Motor execution and motor imagery: A comparison of functional connectivity patterns based on graph theory," *Neuroscience*, vol. 261, pp. 184–194, Mar. 2014.
- [73] R. Bauer, M. Fels, M. Vukelić, U. Ziemann, and A. Gharabaghi, "Bridging the gap between motor imagery and motor execution with a brain–robot interface," *NeuroImage*, vol. 108, pp. 319–327, Mar. 2015.
- [74] G. Schalk, D. J. McFarland, T. Hinterberger, N. Birbaumer, and J. R. Wolpaw, "BCI2000: A general-purpose brain-computer interface (BCI) system," *IEEE Trans. Biomed. Eng.*, vol. 51, no. 6, pp. 1034–1043, Jun. 2004.

- [75] A. L. Goldberger et al., "PhysioBank, PhysioToolkit, and PhysioNet: Components of a new research resource for complex physiologic signals," *Circulation*, vol. 101, no. 23, pp. e215–e220, Jun. 2000.
- [76] (2009). Eeg Motor Movement/imagery Dataset. Accessed: Nov. 1, 2024.[Online]. Available: https://physionet.org/content/eegmmidb/1.0.0
- [77] P. Sale and M. Franceschini, "Action observation and mirror neuron network: A tool for motor stroke rehabilitation," *Eur. J. Phys. Rehabil. Med.*, vol. 48, no. 2, pp. 313–318, Jun. 2012.
- [78] B. Calvo-Merino, D. E. Glaser, J. Grèzes, R. E. Passingham, and P. Haggard, "Action observation and acquired motor skills: An fMRI study with expert dancers," *Cerebral Cortex*, vol. 15, no. 8, pp. 1243–1249, Aug. 2005.
- [79] H. Shibasaki and M. Hallett, "What is the bereitschaftspotential?" Clin. Neurophysiol., vol. 117, no. 11, pp. 2341–2356, Nov. 2006.
- [80] L. Deecke, "Bereitschaftspotential as an indicator of movement preparation in supplementary motor area and motor cortex," in *Proc. Ciba Found. Symp. 132- Motor Areas Cerebral Cortex, Motor Areas Cerebral Cortex, Ciba Found. Symp. 132*, Sep. 2007, pp. 231–250.
- [81] G. Pfurtscheller and F. H. L. da Silva, "Event-related EEG/MEG synchronization and desynchronization: Basic principles," *Clin. Neurophysiol.*, vol. 110, no. 11, pp. 1842–1857, Nov. 1999.
- [82] A. Eliseyev et al., "Recursive exponentially weighted N-way partial least squares regression with recursive-validation of hyper-parameters in brain–computer interface applications," *Sci. Rep.*, vol. 7, no. 1, pp. 1–15, 2017.
- [83] M.-C. Schaeffer and T. Aksenova, "Switching Markov decoders for asynchronous trajectory reconstruction from ECOG signals in monkeys for BCI applications," *J. Physiol. Paris*, vol. 110, no. 4, pp. 348–360, 2016.
- [84] J. L. Collinger et al., "High-performance neuroprosthetic control by an individual with tetraplegia," *Lancet*, vol. 381, no. 9866, pp. 557–564, Feb. 2013.
- [85] Z. Khademi, F. Ebrahimi, and H. M. Kordy, "A review of critical challenges in MI-BCI: From conventional to deep learning methods," *J. Neurosci. Methods*, vol. 383, Jan. 2023, Art. no. 109736.
- [86] G. Huang, Z. Liu, L. Van Der Maaten, and K. Q. Weinberger, "Densely connected convolutional networks," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, May 2017, pp. 4700–4708.
- [87] A. Myrden and T. Chau, "Effects of user mental state on EEG-BCI performance," *Frontiers Hum. Neurosci.*, vol. 9, p. 308, Jun. 2015.
- [88] C. Phunruangsakao, D. Achanccaray, and M. Hayashibe, "Deep adversarial domain adaptation with few-shot learning for motorimagery brain-computer interface," *IEEE Access*, vol. 10, pp. 57255–57265, 2022.
- [89] D. Wu, Y. Xu, and B.-L. Lu, "Transfer learning for EEG-based braincomputer interfaces: A review of progress made since 2016," *IEEE Trans. Cognit. Develop. Syst.*, vol. 14, no. 1, pp. 4–19, Mar. 2022.
- [90] M. Liu and J. Ushiba, "Brain-machine interface (BMI)-based neurorehabilitation for post-stroke upper limb paralysis," *Keio J. Med.*, vol. 71, no. 4, pp. 82–92, 2022.
- [91] J. Leon-Carrion, J. F. Martin-Rodriguez, J. Damas-Lopez, J. M. B. Y. Martin, and M. R. Dominguez-Morales, "Delta–alpha ratio correlates with level of recovery after neurorehabilitation in patients with acquired brain injury," *Clin. Neurophysiol.*, vol. 120, no. 6, pp. 1039–1045, Jun. 2009.
- [92] J. I. Doerrfuss, T. Kilic, M. Ahmadi, M. Holtkamp, and J. E. Weber, "Quantitative and qualitative EEG as a prediction tool for outcome and complications in acute stroke patients," *Clin. EEG Neurosci.*, vol. 51, no. 2, pp. 121–129, Mar. 2020.
- [93] A. Agius Anastasi, O. Falzon, K. Camilleri, M. Vella, and R. Muscat, "Brain symmetry index in healthy and stroke patients for assessment and prognosis," *Stroke Res. Treatment*, vol. 2017, Apr. 2017, Art. no. 8276136.
- [94] G. Pfurtscheller, P. Linortner, R. Winkler, G. Korisek, and G. Müller-Putz, "Discrimination of motor imagery-induced EEG patterns in patients with complete spinal cord injury," *Comput. Intell. Neurosci.*, vol. 2009, no. 1, Jan. 2009, Art. no. 104180.
- [95] K. Amemiya, T. Ishizu, T. Ayabe, and S. Kojima, "Effects of motor imagery on intermanual transfer: A near-infrared spectroscopy and behavioural study," *Brain Res.*, vol. 1343, pp. 93–103, Jul. 2010.
- [96] A. Yokoi, W. Bai, and J. Diedrichsen, "Restricted transfer of learning between unimanual and bimanual finger sequences," *J. Neurophysiol.*, vol. 117, no. 3, pp. 1043–1051, Mar. 2017.
- [97] P. Voss, M. E. Thomas, J. M. Cisneros-Franco, and É. D. Villers-Sidani, "Dynamic brains and the changing rules of neuroplasticity: Implications for learning and recovery," *Frontiers Psychol.*, vol. 8, Oct. 2017, Art. no. 274878.

- [98] V. Zimmern, "Why brain criticality is clinically relevant: A scoping review," *Frontiers Neural Circuits*, vol. 14, Aug. 2020, Art. no. 565335.
- [99] T. Settgast, F. Zilio, A. Kübler, and G. Northoff, "Correlation between neurophysiological measures of consciousness and BCI performance in a locked-in patient," in *Proc. 11th Int. Winter Conf. Brain–Comput. Interface (BCI)*, Feb. 2023, pp. 1–6.
- [100] R. P. Rocha et al., "Recovery of neural dynamics criticality in personalized whole-brain models of stroke," *Nature Commun.*, vol. 13, no. 1, p. 3683, Jun. 2022.
- [101] D. T. Bundy and E. C. Leuthardt, "The cortical physiology of ipsilateral limb movements," *Trends Neurosci.*, vol. 42, no. 11, pp. 825–839, Nov. 2019.
- [102] Y. Gu, O. F. D. Nascimento, M.-F. Lucas, and D. Farina, "Identification of task parameters from movement-related cortical potentials," *Med. Biol. Eng. Comput.*, vol. 47, no. 12, pp. 1257–1264, Dec. 2009.
- [103] T. Hanakawa, I. Immisch, K. Toma, M. A. Dimyan, P. Van Gelderen, and M. Hallett, "Functional properties of brain areas associated with motor execution and imagery," *J. Neurophysiol.*, vol. 89, no. 2, pp. 989–1002, Feb. 2003.
- [104] T. Hanakawa, M. A. Dimyan, and M. Hallett, "Motor planning, imagery, and execution in the distributed motor network: A timecourse study with functional MRI," *Cerebral Cortex*, vol. 18, no. 12, pp. 2775–2788, Dec. 2008.
- [105] S. Bhattacharyya, A. Konar, and D. N. Tibarewala, "Motor imagery and error related potential induced position control of a robotic arm," *IEEE/CAA J. Autom. Sinica*, vol. 4, no. 4, pp. 639–650, Apr. 2017.
- [106] Y. Nishimura, S. I. Perlmutter, and E. E. Fetz, "Restoration of upper limb movement via artificial corticospinal and musculospinal connections in a monkey with spinal cord injury," *Frontiers Neural Circuits*, vol. 7, p. 57, Sep. 2013.
- [107] H. Lorach et al., "Walking naturally after spinal cord injury using a brain-spine interface," *Nature*, vol. 126, pp. 126–133, May 2023.
- [108] F. B. Wagner et al., "Targeted neurotechnology restores walking in humans with spinal cord injury," *Nature*, vol. 563, no. 7729, pp. 65–71, Nov. 2018.
- [109] S. Samejima et al., "Brain–computer-spinal interface restores upper limb function after spinal cord injury," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 29, pp. 1233–1242, 2021.
- [110] N. Kinany, E. Pirondini, R. Martuzzi, L. Mattera, S. Micera, and D. Van De Ville, "Functional imaging of rostrocaudal spinal activity during upper limb motor tasks," *NeuroImage*, vol. 200, pp. 590–600, Oct. 2019.
- [111] K. Zhang, Y. Long, and X. Luo, "Review of supernumerary robotic limbs," J. Phys., Conf. Ser., vol. 2456, no. 1, Mar. 2023, Art. no. 012004.
- [112] C. I. Penaloza and S. Nishio, "BMI control of a third arm for multitasking," Sci. Robot., vol. 3, no. 20, Jul. 2018, Art. no. eaat1228.
- [113] G. Dominijanni et al., "The neural resource allocation problem when enhancing human bodies with extra robotic limbs," *Nature Mach. Intell.*, vol. 3, no. 10, pp. 850–860, Oct. 2021.
- [114] G. Dominijanni et al., "Human motor augmentation with an extra robotic arm without functional interference," *Sci. Robot.*, vol. 8, no. 85, Dec. 2023, Art. no. eadh1438.
- [115] P. Kieliba, D. Clode, R. O. Maimon-Mor, and T. R. Makin, "Robotic hand augmentation drives changes in neural body representation," *Sci. Robot.*, vol. 6, no. 54, May 2021, Art. no. eabd7935.
- [116] K. Arai et al., "Embodiment of supernumerary robotic limbs in virtual reality," *Sci. Rep.*, vol. 12, no. 1, p. 9769, Jun. 2022.
- [117] K. Umezawa, Y. Suzuki, G. Ganesh, and Y. Miyawaki, "Bodily ownership of an independent supernumerary limb: An exploratory study," *Sci. Rep.*, vol. 12, no. 1, p. 2339, Feb. 2022.
- [118] C. Brunner, B. Z. Allison, C. Altstätter, and C. Neuper, "A comparison of three brain–computer interfaces based on event-related desynchronization, steady state visual evoked potentials, or a hybrid approach using both signals," *J. Neural Eng.*, vol. 8, no. 2, Apr. 2011, Art. no. 025010.
- [119] D. Kim and S.-B. Cho, "A brain-computer interface for shared vehicle control on TORCS car racing game," in *Proc. 10th Int. Conf. Natural Comput. (ICNC)*, Aug. 2014, pp. 550–555.
- [120] R. K. Almajidy, Y. Boudria, U. G. Hofmann, W. Besio, and K. Mankodiya, "Multimodal 2D brain computer interface," in *Proc.* 37th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC), Aug. 2015, pp. 1067–1070.

285

- [121] Y. Song, J. Liu, Q. Gao, and M. Liu, "A quadrotor helicopter control system based on brain-computer interface," in *Proc. IEEE Int. Conf. Mechatronics Autom. (ICMA)*, Aug. 2015, pp. 1478–1483.
- [122] D. Furman, R. Reichart, and H. Pratt, "Finger flexion imagery: EEG classification through physiologically-inspired feature extraction and hierarchical voting," in *Proc. 4th Int. Winter Conf. Brain–Comput. Interface (BCI)*, Feb. 2016, pp. 1–4.
- [123] H. Choi, D. P. Jang, and K.-M. Lee, "Bimanual arm movements decoding using hybrid method," in *Proc. 5th Int. Winter Conf. Brain-Comput. Interface (BCI)*, Jan. 2017, pp. 60–62.
- [124] Z. Li, W. Yuan, S. Zhao, Z. Yu, Y. Kang, and C. L. P. Chen, "Brainactuated control of dual-arm robot manipulation with relative motion," *IEEE Trans. Cognit. Develop. Syst.*, vol. 11, no. 1, pp. 51–62, Mar. 2019.
- [125] H. Wang and A. Bezerianos, "Brain-controlled wheelchair controlled by sustained and brief motor imagery BCIs," *Electron. Lett.*, vol. 53, no. 17, pp. 1178–1180, 2017.
- [126] Z. Yijie et al., "A multiuser collaborative strategy for MI-BCI system," in *Proc. IEEE 23rd Int. Conf. Digit. Signal Process. (DSP)*, Nov. 2018, pp. 1–5.
- [127] E. A. Mohamed, M. Z. Yusoff, I. K. Adam, E. A. Hamid, F. Al-Shargie, and M. Muzammel, "Enhancing EEG signals in brain computer interface using intrinsic time-scale decomposition," *J. Phys., Conf. Ser.*, vol. 1123, Nov. 2018, Art. no. 012004.
- [128] A. Reust, J. Desai, and L. Gomez, "Extracting motor imagery features to control two robotic hands," in *Proc. IEEE Int. Symp. Signal Process. Inf. Technol. (ISSPIT)*, Dec. 2018, pp. 118–122.
- [129] R. Takahashi and H. Tanaka, "Motor imagery multi-task classification method," in *Proc. Int. Symp. Affect. Sci. Eng. (ISASE)*, Jan. 2018, pp. 1–6.
- [130] H. Choi, J. Lee, J. Park, B. H. Cho, K.-M. Lee, and D. P. Jang, "Movement state classification for bimanual bci from non-human primate's epidural ecog using three-dimensional convolutional neural network," in *Proc. 6th Int. Conf. Brain–Comput. Interface (BCI)*, 2018, pp. 1–3.
- [131] A. N. Belkacem, S. Nishio, T. Suzuki, H. Ishiguro, and M. Hirata, "Neuromagnetic geminoid control by BCI based on four bilateral hand movements," in *Proc. IEEE Int. Conf. Syst., Man, Cybern. (SMC)*, Oct. 2018, pp. 524–527.

- [132] A. N. Belkacem, H. Ishiguro, S. Nishio, M. Hirata, and T. Suzuki, "Real-time MEG-based brain-geminoid control using single-trial SVM classification," in *Proc. 3rd Int. Conf. Adv. Robot. Mechatronics* (*ICARM*), Jul. 2018, pp. 679–684.
- [133] N. Shajil, S. Mohan, P. Srinivasan, J. Arivudaiyanambi, and A. A. Murrugesan, "Multiclass classification of spatially filtered motor imagery EEG signals using convolutional neural network for BCI based applications," *J. Med. Biol. Eng.*, vol. 40, no. 5, pp. 663–672, Oct. 2020.
- [134] D. Suma, J. Meng, B. J. Edelman, and B. He, "Spatial-temporal aspects of continuous EEG-based neurorobotic control," *J. Neural Eng.*, vol. 17, no. 6, Dec. 2020, Art. no. 066006.
- [135] B. Orset, K. Lee, R. Chavarriaga, and J. D. R. Millán, "User adaptation to closed-loop decoding of motor imagery termination," *IEEE Trans. Biomed. Eng.*, vol. 68, no. 1, pp. 3–10, Jan. 2021.
- [136] S. Tortora, A. Gottardi, E. Menegatti, and L. Tonin, "Continuous teleoperation of a robotic manipulator via brain-machine interface with shared control," in *Proc. IEEE 27th Int. Conf. Emerg. Technol. Factory Autom. (ETFA)*, Sep. 2022, pp. 1–8.
- [137] R. Gao, Y. Liu, J. Wang, and L. Bi, "Multi-direction decoding of bothhand movement using EEG signals," in *Proc. IEEE Int. Conf. Real-time Comput. Robot. (RCAR)*, Jul. 2022, pp. 644–647.
- [138] K. Wang, F. Tian, M. Xu, S. Zhang, L. Xu, and D. Ming, "Restingstate EEG in alpha rhythm may be indicative of the performance of motor imagery-based brain–computer interface," *Entropy*, vol. 24, no. 11, p. 1556, Oct. 2022.
- [139] N. Mammone, C. Ieracitano, H. Adeli, and F. C. Morabito, "AutoEncoder filter bank common spatial patterns to decode motor imagery from EEG," *IEEE J. Biomed. Health Inf.*, vol. 27, no. 5, pp. 2365–2376, May 2023.
- [140] Z. Shuqfa, A. N. Belkacem, and A. Lakas, "Decoding multi-class motor imagery and motor execution tasks using Riemannian geometry algorithms on large EEG datasets," *Sensors*, vol. 23, no. 11, p. 5051, May 2023.
- [141] H. L. Reynolds, M. A. Hanafy, J. Jones, D. Harris, and D. O. Popa, "Comparative study of one-handed vs two-handed EEG intent recognition for applications in human-robot interaction," in *Proc. 17th Int. Conf. Pervasive Technol. Rel. Assistive Environ.*, Jun. 2024, pp. 496–503.